%  Midterm examination rules
%  Combinatorial Optimization, SUAMI, summer 2015
%  Brian Kell <bkell@cmu.edu>

\pdfpagewidth=8.5truein \pdfpageheight=11truein
\hsize=5.5truein \vsize=9truein
\pdfhorigin=1.5truein \pdfvorigin=1truein

\font\titlefont=cmbx12

\headline={\hfil}
\footline={\hfil}

\edef\restoreinterlineskip{\noexpand\baselineskip=\the\baselineskip\relax\noexpand\lineskiplimit=\the\lineskiplimit\relax\noexpand\lineskip=\the\lineskip}

\long\def\boxed#1{\vbox{\offinterlineskip
	\hrule
	\hbox to \hsize{%
	\hss\vrule\hskip1em%
	\vbox{\advance\hsize by-2em \advance\hsize by-0.8pt
	\restoreinterlineskip
	\vskip1ex
	\noindent#1\par
	\vskip1ex
	}\hskip1em\vrule\hss}%
	\hrule}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\centerline{\titlefont Combinatorial Optimization}
\smallskip
\centerline{\bf Midterm examination rules}

\medskip

\noindent Please read all of the following before opening the envelope
containing the questions.

\bigskip

\input midterm-rules-include.tex

\bigskip

\noindent After completing the exam, please sign the following statement and
submit it with your solutions.

\medskip

\boxed{%
\noindent I have read, understood, and followed all of the rules for this
examination. On my honor, I have neither received nor given any unauthorized
help, and all of the work in my solutions is my own.

\vskip0.6in

\hrule\vskip2pt
\hbox to\hsize{\strut Signature\hfill Date\hskip1in}
}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\bye
